Author's response to reviews

Title: Diagnostic yield of transbronchial cryobiopsy in non-neoplastic lung disease: a retrospective case series

Authors:

Sergej Griff (sergej.griff@helios-kliniken.de)
Nicolas Schönfeld (nicolas.schoenfeld@helios-kliniken.de)
Wilhelm Ammenwerth (wim@ammenwerth.com)
Torsten-Gerriet Blum (torsten-gerriet.blum@helios-kliniken.de)
Christian Grah (christian.grah@havelhoehe.de)
Torsten T. Bauer (torsten.bauer@helios-kliniken.de)
Wolfram Grüning (wolfram.gruening@helios-kliniken.de)
Thomas Mairinger (thomas.mairinger@helios-kliniken.de)
Henrik Wurps (henrik.wurps@helios-kliniken.de)

Version: 3 Date: 11 October 2014

Author's response to reviews: see over
Berlin, 11.10.2014

Dear Editors of BMC Pulmonary Medicine,

after having received your email on October 2, 2014, we have again revised our manuscript and have integrated the following modifications:

- The kind of (statistical) study was added to the title
- The email addresses of all authors were added on the title page
- The sections “competing interests” and “key words” were replaced
- The abstract was divided into four sections (background, methods, results, conclusion)
- The order and proper names of all sections were rearranged according to BMC guidelines
- The statistics section was integrated into the methods section
- The figures itself were deleted from the main manuscript, only the figure legends were left here
- The table was modified according to BMC guidelines

We hope that all concerns are now solved and that you will consider our newly revised manuscript.

Best regards,

Sergej Griff and Henrik Wurps.